Boston University School of Law

Scholarly Commons at Boston University School of Law

Faculty Scholarship

5-2024

Marketing, Other Intangibles, and Output Growth in 61 United States Industries

Leo Sveikauskas

Rachel Soloveichik

Corby Garner

Peter B. Meyer

James Bessen Boston University School of Law

See next page for additional authors

Follow this and additional works at: https://scholarship.law.bu.edu/faculty_scholarship

Part of the Law and Economics Commons, and the Marketing Law Commons

Recommended Citation

Leo Sveikauskas, Rachel Soloveichik, Corby Garner, Peter B. Meyer, James Bessen & Matthew Russell, *Marketing, Other Intangibles, and Output Growth in 61 United States Industries*, *in* Review of Income and Wealth (2024).

Available at: https://scholarship.law.bu.edu/faculty_scholarship/3966

This Article is brought to you for free and open access by Scholarly Commons at Boston University School of Law. It has been accepted for inclusion in Faculty Scholarship by an authorized administrator of Scholarly Commons at Boston University School of Law. For more information, please contact lawlessa@bu.edu.



Authors

Leo Sveikauskas, Rachel Soloveichik, Corby Garner, Peter B. Meyer, James Bessen, and Matthew Russell



Review of Income and Wealth Series 0, Number 0, 2024 DOI: 10.1111/roiw.12678

MARKETING, OTHER INTANGIBLES, AND OUTPUT GROWTH IN 61 UNITED STATES INDUSTRIES

BY LEO SVEIKAUSKAS*

The Bureau of Labor Satistics

RACHEL SOLOVEICHIK

The Bureau of Economic Analysis

CORBY GARNER

The Bureau of Labor Satistics

PETER B. MEYER

The Bureau of Labor Satistics

JAMES BESSEN

The Technology & Policy Research Initiative program at the Boston University School of Law

AND

MATTHEW RUSSELL

The Bureau of Labor Satistics

Experts in the System of National Accounts (SNA) recently considered whether marketing could be included as a capital asset in the national accounts and later recommended that marketing should be an intangible in the 2025 SNA (IMF, 2022, 2023). This paper prepares macroeconomic measures of the United States marketing stock and develops similar measures within 61 industries. We find that, from 1987 to 2020, marketing capital contributed approximately as much to output growth (0.18 percentage point per year) as R&D (0.15) or software (0.19) did. Software grew more rapidly, but marketing had a larger factor share. Marketing contributes even more to output growth if quality is adjusted to allow for the better targeting associated with digital advertising. There is a close relationship between data flows, software, and digital marketing and national accountants will have to allocate expenditures among these categories.

JEL Codes: M31, M37, O63

Keywords: marketing, advertising, intangibles, intellectual property capital

Sveikauskas and Meyer are research economists and Garner and Russell are economists at the Bureau of Labor Statistics. Soloveichik is a research economist at the Bureau of Economic Analysis. Bessen leads the Technology & Policy Research Initiative program at the Boston University School of Law. The authors thank Richard Graham, Thomas Howells, Leonard Nakamura, Sabrina Pabilonia, Greg Prunchak, Jon Samuels, Jay Stewart, and anonymous referees for their helpful comments and advice. This paper represents the views of the authors and does not necessarily reflect those of the Bureau of Labor Statistics or the Bureau of Economic Analysis.

*Correspondence to: Leo Sveikauskas, The Bureau of Labor Satistics, Washington, D.C., USA (sveikauskas.leo@bls.gov).

Published 2024. This article is a U.S. Government work and is in the public domain in the USA. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

1. INTRODUCTION

In recent years, intangibles have played an increasing role in discussions of economic growth. The early study by Corrado et al. (2005) was especially influential because it established the framework within which economists typically examine the importance of intangibles. Subsequent work has improved measurement and understanding of many intangible assets. Corrado et al. (2009) concluded that incorporating intangibles in national accounts substantially increased measures of capital deepening and somewhat raised labor productivity growth.

Empirical research has shown that marketing often increases purchases for several years and therefore qualifies to be counted as investment. An early experiment demonstrated that random adjustments in the amount of advertising on cable television affected household purchases of products for at least two years (Lodish et al., 1995). More recent research used natural experiments to show that advertising influences behavior for years (Bronnenberg et al., 2012; Bursztyn & Cantoni, 2016).

Corrado & Hao (2014) prepared comprehensive estimates of marketing investment for the U.S. macroeconomy, combining estimates of purchased advertising, several additional types of purchased marketing services, and own-account marketing. Heys & Fotopoulou (2022) consider investment in design, organizational capital, firm-specific training, branding, and financial product innovation. Corrado et al. (2022a) conclude that economic researchers should include the full complement of intangibles.

Statistical agencies have been slower to bring intangibles into official statistics. The System of National Accounts (SNA) now includes software, research and development (R&D), and entertainment originals as investment.¹ The SNA has recently considered including marketing assets as an additional type of intangible investment.² As part of that discussion, IMF (2022) requested comments on capitalization of marketing assets and, in response, we identify and discuss a number of relevant issues. A subsequent document (International Monetary Fund [IMF], 2023) concluded that marketing should be a further intangible in the 2025 SNA and asked for "conceptual and practical guidance … to implement this recommendation." We think that many elements of our paper will be useful in implementation.

This paper develops macroeconomic measures of marketing assets broadly similar to Corrado & Hao (2014) and Heys & Fotopoulou (2022). We also construct and analyze measures of marketing investment for each of our 61 industries that jointly comprise the U.S. private business sector.

Our measures of marketing are based on input-output (IO) tables and occupational information. First, we obtain data on each industry's purchases of advertising from the U.S. IO tables; purchased advertising is defined as the

¹"Entertainment, literary, and artistic originals" is an expression used in the SNA. The U.S. Bureau of Economic Analysis refers to this category as "entertainment originals." We use that terminology throughout this paper.

²The 2022 IMF document is available at https://unstats.un.org/unsd/nationalaccount/SNAUpdate /GZTT.asp. The draft cited here is from row G.9 in the column marked Endorsed IMF (2022). The March 2023 document that supported marketing as an investment is IMF (2023).

commodity associated with North American Industry Classification System (NAICS) industry 5418 (advertising, public relations, and related services).³ Second, we measure each industry's purchases of other marketing services by its purchases from selected portions of NAICS industries 5182, 5415, 5416, and 5419, again from the IO tables. Third, we develop stocks of own-account marketing from occupational data collected by the U.S. Bureau of Labor Statistics (BLS) Occupational Employment and Wage Statistics (OEWS). We follow Corrado & Hao (2014) and Heys & Fotopoulou (2022) in converting measures of occupations into own-account stocks. By combining data from these three sources, we develop measures of marketing assets for the U.S. private economy and for each industry. The rest of this paper refers to these joint measures of purchased advertising, other purchased marketing services, and own-account marketing as marketing.

Our work contributes to ongoing discussion along two main lines. First, we develop prototype measures of the extent and impact of marketing investment in the United States. These measures cover the U.S. private economy and each industry. Our analysis shows that it is feasible to develop reasonable measures of marketing assets for the United States. The paper also considers several potential difficulties that statistical agencies will have to address as they measure marketing. Second, we use information on marketing and other intangibles to examine sources of growth in various industries.

Section 2 below outlines the framework within which this study is conducted. Section 3 describes how we develop measures of purchased and own-account marketing, which are the central ingredients of our study. Section 4 considers how these new measures of marketing investment affect United States macroeconomic growth. This section also compares the overall contribution of marketing with the impact of other sources of growth. Section 5 uses detailed industry data to examine several issues about marketing. Section 6 examines the relationships between marketing, other intangibles, and additional sources of growth within data for individual industries. Section 7 concludes. The Appendices provide further information on how we calculate stocks of purchased and own-account marketing and measure their impact on the economy.

2. Theoretical Framework

As in many studies of intangibles, we measure output by value added in constant dollars. Capital services are measured by quantities of assets weighted by their corresponding rental prices. Labor is measured in hours. We begin with a production function, expressed in growth rates:

³The intuition behind our approach is expressed most clearly in terms of the IO tables. In practice we use a slightly more complex procedure. The complexity arises because we wish to make sure that our measures of purchased marketing and our data on own-account marketing both contain similar elements of marketing. In order to match the occupations that we have data for, we remove conventions and trade shows, which are ordinarily included in the commodity advertising, from our advertising data and, conversely, add further data on signs that are not normally contained in the commodity advertising. Similarly, when we use data from the Economic Census or the Service Annual Survey to measure the presence of marketing as a proportion of output provided by each industry, we follow the IO tables by adjusting the Census data for the well-known issues associated with underreporting or misreporting of incomes (U.S. Bureau of Economic Analysis, 2021, pp. 11–14).

(1)
$$v_{j,t} = (\alpha_k)_{j,t} k_{j,t} + (\alpha_l)_{j,t} l_{j,t} + tfp_{j,t}$$

where $v_{j,t}$ is the rate of growth of real value added in industry *j* in year *t*, $k_{j,t}$ is the rate of growth of capital service input, and $l_{j,t}$ is the rate of growth of labor input. $tfp_{j,t}$ is the corresponding growth of total factor productivity, typically calculated as a residual. $(\alpha_k)_{j,t}$ and $(\alpha_l)_{j,t}$ are the cost shares for capital and labor, each calculated as averages for years *t* and t - 1.⁴

The effect that any specific capital service, *i*, has on output growth follows the framework implied in expression (1). Specifically:

(2)
$$v_{i,j,t} = \left(\alpha_k\right)_{i,j,t} k_{i,j,t}$$

where $(\alpha_k)_{i,j,t}$ is the share of asset *i* in the value added of industry *j* in year *t*. $k_{i,j,t}$ is correspondingly the growth of service *i* in that same industry and year. The longer-term contribution of any capital service to output growth for the 33 years from 1987 to 2020, $LTCON_{i,j}$, is similarly:

(3)
$$LTCON_{i,j} = \left\{ \left[\prod_{t=1988}^{2020} \left(v_{i,j,t} + 1.00 \right) \right] \right]^{\left(\frac{1.0}{33.0} \right)} \right\} - 1$$

as calculated from the geometric mean of one plus the annual contributions.⁵

Our study considers seven different types of intangibles: R&D, entertainment originals, own-account software, custom software, pre-packaged software, purchased marketing, and own-account marketing. Each intangible is studied in 61 industries over the 1987–2020 period. To remove the effects of business cycles, we present results for the 1990–2000, 2000–2007, and 2007–2020 subperiods.⁶ We often measure the relative importance of different forms of capital through their shares of capital services and their contributions to output.

3. STOCKS OF PURCHASED AND OWN-ACCOUNT MARKETING

3.1. Stocks of purchased advertising

As Corrado et al. (2009, p. 670) remark, "Expenditures for advertising are a large part of the investments in brand equity." Purchased advertising is the largest

⁴Our measures of capital stocks and services are calculated as Tornqvist indexes, using BLS Productivity Program methods. Labor composition indexes are also prepared with Tornqvist aggregation. The BLS obtains value added output data from the Bureau of Economic Analysis; these data are based on Fisher indexes.

⁵Stiroh (2002, equation (7), page 1172) analyzes how industries contribute to national labor productivity growth in a value-added framework. In contrast, this paper concentrates on contributions to growth within industries.

⁶United States recessions begin in July 1990, March 2001, December 2007, and February 2020. We select 1990 and 2007 as initial points in which the economy was still growing for a considerable part of the year in question, and 2000 as the last normal year prior to March 2001. We extend the 2007–2019 period to include 2020 because the COVID–19 recession was brief.

single element of marketing that we consider in this study. We measure how much advertising each industry acquires by its purchases of the commodity "advertising."⁷ This includes advertising purchased from NAICS industry 5418, "Advertising, public relations, and related services," as well as advertising purchased from other industries such as print media, radio and TV, and the internet. We work with the commodity version of purchased advertising because the commodity data include all advertising that each industry purchases regardless of its source.

We use the IO tables to estimate industry purchases of advertising and other sources of purchased marketing services. For 1997 to 2020, we use the annual IO use tables developed by the Employment Projections program of the BLS. For 1982 to 1996, we use the U.S. Bureau of Economic Analysis (BEA) Historical Input–Output Tables, which offer less industry detail. We calculate the ratio of "advertising, public relations, and related services" to "miscellaneous professional, scientific, and technical services" in each industry in 1997 and use each industry-specific ratio to approximate advertising expenditures from 1982 to 1996. Our assumptions concerning depreciation imply that investments made prior to 1982 have fully depreciated by the time our analysis begins in 1987.

There has been some controversy about the usefulness of IO information to measure advertising, both at the individual industry level (Rogers & Tokle, 1995) and at the aggregate level (Silk & Berndt, 2020). To illustrate how the IO commodity data measure aggregate advertising, consider data for the year 2012. Silk & Berndt (2020, p. 47) suggest that, in 2012, firms that supply advertising and marketing services, such as ad agencies, had receipts of approximately \$90 billion, and that providers of media access, such as broadcasters or print and internet providers, had about an additional \$180 billion in revenue, implying total expenditures of approximately \$270 billion. The graph in Figure 3. of their paper suggests that advertising expenditures reported to the IRS were perhaps a little closer to \$280 billion. The data used in this paper imply that at the commodity level advertising expenditures in the private economy were approximately \$305 billion in 2012.

To deflate advertising expenditures, for 1997 to 2020 we use the BEA price index for the gross output price of commodities in NAICS industry 5418 ("advertising, public relations, and related services").⁸ This BEA price deflator incorporates Producer Price Indexes (PPIs) for internet publishers, newspapers, radio, and TV, and other industries that produce advertising and also reflects certain other costs. For years prior to 1997 we prepare a new commodity price index that also reflects PPIs and certain costs.⁹ Appendix S3 briefly describes how we prepared prices for 1982 to 1997. We use the price index for advertising to measure the price of output for every form of marketing.

⁷Corrado and Hao (2014, pp. 21–24) present several reasons why they believe that advertising has an effect at the industry level and is not simply dissipated by marketing expenditures of rivals. The official guidelines for national accounting exclude externalities from the national accounts (United Nations Statistical Division, 2009, Section 3.92) so long-lived marketing would be capitalized even if it were canceled out by competitors' advertising.

⁸The BEA deflator for advertising is from the "Gross Output by Industry" files, under Underlying Detail, tab UGO 204–A, table line 144. We do not use the Producer Price Index for "Advertising space and time sales" (WPU 36) because that does not cover all advertising services or years before 2009.

⁹Prior to 1997 the PPI provides much less information about the price of services.

There is some question as to how well existing price deflators measure the output price of marketing. Mandel (2019) argues that the quality-adjusted price of advertising has declined rapidly in recent years because digital advertising is more effective than previous marketing methods. In particular, digital advertising can target potential customers more precisely than print or broadcast advertising can. Section 5.3 considers Mandel's important hypothesis in more detail.

The question of what percentage of advertising expenditures represents investment is a central issue on which there is little conclusive evidence. We therefore adopt the same investment ratios used in other studies. The U.K. Office of National Statistics (ONS) has been a leader in the analysis of intangibles. Heys & Fotopoulou (2022), of the ONS, assume that 60 percent of purchased advertising services and 80 percent of purchased marketing services represent investment. We adopt these percentages in our baseline measures. Our alternative measure follows Corrado et al. (2005, 2009) and Corrado & Hao (2014) and assumes that 60 percent of purchased advertising services represents investment.

On the basis of Corrado & Hao (2014), Villalonga (2004), and Corrado et al. (2009), we assume a 45 percent depreciation rate for our main analysis and 65 percent as an alternative depreciation rate. These rates imply service lives of 4 and 2 years, respectively.¹⁰ We use these same rates of depreciation for all forms of marketing. Once we have determined nominal expenditures, the deflator, the proportion of expenditures that is investment, and depreciation, we measure stocks of each asset through standard perpetual inventory calculations.

3.2. Purchases of other marketing services

Firms purchase marketing services from industries other than advertising (NAICS 5418). Corrado & Hao (2014) include purchases from marketing consulting (NAICS 541613) and market research (NAICS 541961). We also include website design and hosting purchased from NAICS industries 5182 and 5415.¹¹ To the best of our knowledge, our study is the first work to include web design and hosting as marketing investment. For NAICS industries 5182, 5416, and 5419, we first calculate the proportion of output from each industry that represents marketing services; we estimate the presence of marketing services from data in the quinquennial Economic Census and then adjust for under- and misreporting. Between Census years, we use the Service Annual Survey (SAS) to interpolate and extrapolate. Such data provide reasonable information on

¹⁰Corrado et al. (2005, 2009) initially assumed 60 percent annual depreciation. Corrado and Hao (2014, p. 10) recommend lengthening the service life of investments in brand from three to about four years, which approximately corresponds to 45 percent depreciation. Vitorino (2014, p. 20) selects a 20 percent depreciation rate for advertising. However, Bagwell (2005, p. 44), using similar information, suggests a greater rate of depreciation. We believe that, on balance, the overall evidence indicates that advertising depreciates more rapidly than 20 percent per year.

¹¹Section 3.1 used the commodity advertising rather than the industry advertising because the overall production of advertising is far greater than output in the advertising industry (5418). In other industries that supply marketing services, commodity output is very close to industry output, and we use the standard industry data.

overall purchases of marketing services, but, as Appendix S3 explains, it is a challenge to assign these amounts to specific purchasing industries. IO tables do not provide sufficient detail to track purchases of very detailed goods. We are therefore forced to allocate purchased marketing services to the industries that use them through data for the next higher level IO sector. Since we include purchases of marketing from additional industries, our estimates of purchased marketing are generally larger than those in Corrado and Hao. Appendix S1 shows how much each type of marketing service contributes to investment in marketing in each year.

Nakamura et al. (2017) suggest that each of these estimates of marketing should be priced at the price of overall advertising. They find that advertising viewership costs are more closely associated with each other than with measures of content creation. Figure 9 of their study shows that the viewership cost of digital media is correlated with viewership costs in other media and that this correlation increased in the 2010s as digital media became more prevalent. For this reason, we use the BEA advertising price index, instead of a cloud price deflator or other content creation costs, to price all forms of marketing purchases. Section 5.3 emphasizes that the topic of adjusting marketing output prices for unmeasured quality change requires further consideration.

Stocks of own-account marketing

The literature typically draws a sharp distinction between purchased marketing and own-account marketing expenditures. While it is useful to know the approximate magnitudes of each of these two types of expenditures, we caution that these expenditures are inevitably closely interrelated. Internal marketing personnel are highly involved in external marketing campaigns. From this perspective, estimates of total marketing are more reliable than separate estimates of purchased and own-account resources. In the final analysis, the total marketing effort is what really counts.¹²

Own-account marketing expenditures are generally measured based on occupational employment.¹³ We use the presence of certain occupations in each industry to measure the quantity of own-account expenditures. We do not distinguish between own-account advertising and marketing but instead define an overall own-account category that we call own-account marketing.

We obtain each industry's occupational employment for 2002 to 2020 from the OEWS.¹⁴ The OEWS is collected over a rotating three-year cycle, in which a third of

¹²It is also difficult to distinguish between own-account production and output that is sold to customers.

¹³Nakamura et al. (2017) also argue that measures of purchased advertising should be supplemented by data on own-account marketing. For example, a television network might use unsold advertising slots to promote an upcoming show. Following Soloveichik (2013) and Nakamura et al. (2017), we also include radio, TV, and other media expenditures that advertise their own product as part of own-account expenditures.

¹⁴The OEWS began to collect some initial data in the 1990s. However, the initial data used different industry codes and sometimes different occupation codes. Methods and classification systems changed over time and the OEWS program does not recalculate older data to make sure that the information

the sample is collected each year. For every occupation-industry pair, we assign each three-year average to the middle year. Appendix S2 lists the occupations that we assigned to marketing and describes how information on occupations is converted into own-account marketing stocks. Before 2002, we extrapolate own-account marketing in each industry with data on aggregate occupational employment from the OEWS, on output of each industry, and on purchased marketing services.

Estimates of the time that each occupation spends on long-term investment would ideally depend on careful time studies. Unfortunately, this type of conclusive evidence does not appear to exist. Our baseline measure follows Heys & Fotopoulou (2022) and assumes that 30 percent of own-account expenditures are investment. Our alternative measure follows Corrado & Hao (2014) and assumes that 60 percent of own-account expenditures are investment. We assume that own-account marketing depreciates at the same 45 percent rate as purchased advertising, with 65 percent as an alternative. Once our assumptions about expenditures, deflators, the investment portion, and depreciation are set, we construct perpetual inventory stocks of own-account marketing for each industry.

Existing work on own-account marketing (Corrado & Hao, 2014; Heys & Fotopoulou, 2022) uses a relatively narrow list of relevant occupations. We think it is possible that a wider range of occupations, especially in sales, may also contribute to the value of marketing assets. Many sales workers develop continuing relationships with their customers that eventually lead to greater long-term sales. We do not know of any empirical studies that document how much time sales workers spend investing in longer term relationships. However, because sales workers are such a large group, even a small proportion of their time could substantially increase measures of marketing investment. We think that this is a potentially important topic that should be carefully considered before marketing is included in the accounts.

The IMF discussion of marketing assets frequently refers to the value of trademarks and logos. Dosi et al. (2022) estimated how much a new trademark, in itself, adds to the value of a firm. However, we believe that the value of a trademark more fundamentally reflects a firm's underlying assets, including its marketing, R&D, and organizational capabilities. We think that future work that integrates the value of a trademark with these underlying capabilities will strengthen the usefulness of measures of marketing assets.

3.3. Adapting existing data to include marketing assets as an additional intangible

The BLS Productivity Database contains many data elements that are useful in measuring the impact of marketing. This includes gross output and value added, in both current and constant dollars, and measures of K (capital), L (labor), E (energy), M (materials), and S (purchased services). The data on E, M, and S together provide measures of intermediate input. This subsection describes how we measure output and input from the BLS data and how we modify existing BLS data to allow for purchased and own-account marketing as additional intangibles.

is consistent over time. Our estimates of the growth of occupational employment and earnings may therefore differ somewhat from a consistent time series.

Published 2024. This article is a U.S. Government work and is in the public domain in the USA. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth.

We begin with the existing measures of gross output and purchased services and the new measures of marketing investment described above. For each of these series we have prices in current dollars and chain-type quantity indexes. In addition, we have measures of value added from BEA. BEA prepares value added by double deflation, deflating both gross output and intermediate inputs (Moyer et al., 2004). The investment portion of marketing must be removed from each industry's purchased services and transferred to capital investment. By construction, a smaller quantity of purchased services requires that intermediate prices be recalculated for each industry. This new price is then used to compute adjusted quantities of intermediate inputs. We use the double deflation method described in Moyer et al. (2004) to remove marketing investment from purchased services, recalculate intermediates, and recalculate value added output by removing our new measures of intermediate input from gross output.

Gross output and value added both increase when portions of marketing are treated as investment. It is necessary to decide where to allocate the extra value-added income. Previous work on intangibles in the U.S. Accounts, such as studies of R&D (Fraumeni & Okubo, 2005) and software, has assumed that the added income from capitalization all goes to capital. To be consistent with those studies, we also assume that the added income from capitalization goes to capital, and that there is no effect on employee compensation.¹⁵

The decision to assign all additional income from capitalization of intangibles to capital has important implications. Koh et al. (2020) show that the decline in the labor share observed in the U.S. occurs solely because all the additional income from intangibles is assigned to capital. They argue that such an allocation is "extreme," and that a portion of the new value added created should instead be assigned to labor; they recommend detailed micro analysis to determine where extra output should be assigned. The Koh et al. study is insightful and thought provoking. If further work supports their interpretation, some of the value created by capitalizing marketing and other intangibles may eventually be assigned to labor, and existing estimates of property income and the associated rental prices are probably too high.

Once we have constructed stocks of purchased and own-account marketing and estimated the increase in property income associated with these investments, we are ready to value these stocks. To determine rental prices, we treat purchased and own-account marketing just like any other capital asset. As is standard procedure, we begin with data on property income in each industry and year, determine an internal rate of return for each industry/year, and then calculate rental prices that reflect asset price changes, rates of depreciation, and tax parameters.¹⁶

¹⁵Our estimates assume that the increased income from marketing investment goes to an increase in the gross operating surplus. The increased surplus goes to corporate profits or to the capital or labor income of proprietors. We use ratios from the existing national accounts to assign the new surplus to each of these three categories in each industry. Almost all the additional surplus is assigned to capital, since proprietor labor income is typically a small share of total gross operating surplus.

¹⁶BLS does not use an internal rate of return for every industry and year, because rates of return calculated from property income are sometimes negative or otherwise implausible, especially in recessions or when the capital gains term is large. In these cases, BLS uses an external rate of return to calculate rental prices. The external rate of return is the average rate of return observed in the private business sector, adjusted for differences in the capital gains term.

4. The Macroeconomic Impact of Marketing Assets

This section analyzes how the new measures of marketing assets affect macroeconomic growth in the private sector. First, we measure the contributions that purchased and own-account marketing, other intangibles, other inputs, and the Total Factor Productivity residual (TFP) make to output growth. Second, we look at the flow of capital services to goods and services industries.

4.1. The effect of marketing on output growth

Before examining how each intangible contributes to output growth, we show some of the main patterns in the overall data. Panel a of Figure 1 shows how intangibles, which now include the new purchased and own-account marketing assets, have consistently grown more rapidly than tangibles. Panel b shows that intangibles, which originally were less influential than information and communication technology capital or other assets, are now more important. This occurred because other forms of capital made less of a contribution, not because the contribution of intangibles increased.

Table 1 reports the central results on how each intangible contributes to output growth. These estimates reflect our preferred (baseline) assumptions concerning the proportion of marketing expenditures that represents investment and the rate of



Figure 1. Panel a: Capital Services Growth of Intangible and Tangible Assets in the Private Economy. *Notes:* Data Show the Average Annual Growth of Capital Services in each Subperiod. Panel b: Contributions of Different Types of Assets to Private Capital Growth. Data Show the Average Annual Percentage Point Contribution that each Category of Assets Contributes to the Growth of Capital Services

	1990-2000	2000-2007	2007-2020	1987-2020
Purchased marketing	0.14	0.16	0.17	0.15
Own account marketing	0.03	0.03	0.05	0.03
Entertainment originals	0.04	0.04	0.02	0.03
Research and development (R&D)	0.17	0.13	0.14	0.15
Software, custom	0.07	0.05	0.07	0.06
Software, own account	0.03	0.04	0.03	0.03
Software, prepackaged	0.11	0.10	0.11	0.10
Information and Communication	0.54	0.39	0.22	0.37
Technology (ICT)				
All other assets	0.56	0.41	0.28	0.42
Labor input	1.43	0.30	0.31	0.75
TFP residual	0.93	1.10	0.45	0.71
Private business output	3.9	2.5	1.7	2.7

TABLE 1
INPUT AND TFP CONTRIBUTIONS TO PRIVATE BUSINESS OUTPUT GROWTH

Notes: Table 1 shows how each input and TFP affect output growth in the private business sector. The estimates in Table 1 are based on our preferred (baseline) assumptions concerning the proportion of each type of expenditure that is classified as investment and the rate of depreciation, as reported in Table C.1. The second portion of Appendix S3, on the impact of capital inputs on output growth, explains how we aggregate capital inputs and measure their impact on output growth.

depreciation. Table 1 shows that, of the presently recognized intangibles, R&D and software have the greatest impact on macroeconomic growth. Over the entire 1987 to 2020 period, R&D contributed 0.15 percent a year to output growth and the three types of software together added 0.19 percent a year. The two types of marketing contributed 0.18 percent a year to output growth. This evidence makes the important point that marketing contributes about as much to output growth as R&D or software do. Appendix S3 provides further detail on how we obtain these results.

Background information helps to clarify the effects of both R&D and software. Table 1 includes only the direct effects of R&D—the immediate returns to firms that initially conduct research. Evidence shows that R&D spillovers account for more than half of the total returns to R&D and that the spillover portion of total returns has increased in recent years (Bloom et al., 2013; Lucking et al., 2019; Sveikauskas, 2007). These well-documented spillovers show that social returns to R&D are much greater than the private returns shown in Table 1. It has so far been difficult to assign R&D spillovers to specific industries. However, Martin et al. (2022) recently developed a framework that may be able to assign R&D spillovers to each industry.

Pre-packaged software affects growth more strongly than other categories of software do, as Table 1 indicates. That might seem to contradict Bessen (2020, 2022), who has shown that large firms often dominate their industries by developing highly productive proprietary computer systems; these powerful proprietary systems might seem to be own-account software. However, BEA classifies software-as-a-service (SaaS) as pre-packaged software, and this category has grown rapidly, so U.S. data show that pre-packaged software contributes strongly to growth.¹⁷

¹⁷BEA specialists mention that some countries prefer to think of software as a purchased service. However, BEA plans to continue classifying SaaS as an investment in pre-packaged software. They also

1987–2020	TFP	Output	Capital	Labor
Private industry without any marketing	0.752	2.707	3.388	1.174
Private industry with only purchased marketing	0.703	2.762	3.599	1.174
Private industry with all marketing	0.699	2.784	3.655	1.174

 Table 2

 Purchased And Own-Account Marketing Long-Run Effects

Notes: The first two lines show the impact on output and inputs when (commodity) advertising, web design and hosting services, marketing consulting, and marketing surveys are capitalized. A comparison of the last two lines reports similar results when own-account marketing is also capitalized.

Despite the importance of marketing, inclusion of marketing as an additional intangible does not greatly increase measured economic growth. Table 2 shows that capitalization of purchased and own-account marketing increases output growth by less than 0.1 percent a year. This growth increase is similar to the increase associated with capitalization of R&D (Ribarsky, 2022).

Table 1 had reported our preferred estimates of the role of marketing, and its importance relative to R&D and software, in the United States economy. However, it is also useful to present supporting information showing corresponding effects under a variety of different assumptions. Table S.2 of Appendix S3 reports results under several alternative assumptions. These sensitivity tests show that changes in the percentage of marketing expenditures that is investment have a considerable impact on the implied contribution of marketing. Since advertising expenditures are substantial, changes in this investment proportion are particularly influential. In contrast, changes in the rate of depreciation, within the range of values that the literature suggests, have little effect on the implied role of marketing. These results suggest that, as further work on marketing proceeds, researchers could usefully concentrate on measuring the proportion of expenditures that represents long-term investment. Time diaries and interviews and surveys of workers might provide more evidence.

Figure 2 shows that investment as a percentage of business value-added output increased for both purchased and own-account marketing in recent years. Figure S.2 in Appendix S1 shows that the investment-to-output ratio similarly increased for most components of marketing since 2010.

Figure 2 also shows that purchased marketing accounts for a considerably larger proportion of investment than own-account marketing does. In part, this pattern arises because our baseline estimates assume that only 30 percent of own-account marketing expenditures represent investment. If we assume instead that 60 percent of own-account expenditures is investment, then the lower line in Figure 2 would be twice as high and much closer to the top line (purchased marketing).

Table 3 shows the rate of growth of investment for various types of capital assets in different time periods. Investment growth slowed over time for most asset categories. Prepackaged software grew rapidly in each time period. It might seem

comment that the increased reliance on cloud computing has altered the way that firms pay for software and that counting SaaS as an investment in software helps to describe the new payment patterns.

Published 2024. This article is a U.S. Government work and is in the public domain in the USA. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth.



Figure 2. Marketing Investment as a Share of Value Added in the Business Sector. *Note:* The Figure Shows Trends in Nominal Purchased and Own-Account Marketing Investment as a Share of Value-Added Output in the Private Business Sector

	,,		
1990-2000	2000-2007	2007-2020	1987-2020
10.7	6.4	4.0	7.3
10.8	7.1	6.8	8.6
4.3	2.5	1.0	2.5
5.5	2.7	3.6	4.2
16.2	4.3	6.1	9.3
7.4	3.2	4.2	5.3
24.0	9.8	11.0	17.8
8.4	6.4	4.5	6.0
3.2	1.3	0.8	1.7
4.5	2.4	2.5	3.1
	1990-2000 10.7 10.8 4.3 5.5 16.2 7.4 24.0 8.4 3.2 4.5	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

TABLE 3 GROWTH OF REAL INVESTMENT BY ASSET CATEGORY, ANNUAL AVERAGE

Notes: Reports the growth of real investment in each form of capital, including each type of intangible.

surprising that prepackaged software has grown so quickly (Table 3), whereas purchased marketing contributed more to output growth (Table 1). Table 4, which shows the annual rates at which stocks grow, the factor shares, and the contributions to output growth for each of these two types of assets, explains these different patterns. Purchased marketing's larger factor share offsets the more rapid growth of pre-packaged software and allows purchased marketing to contribute more to growth.

4.2. The flow of capital services in goods and services

The goods sector consists of agriculture, mining, utilities, construction, and manufacturing. Services are the rest of the private economy. The private economy

	Growth or percent of	f stocks, change	Factor shar	e, percent	Contribu output grow	tion to th, percent
	Software, prepackaged	Purchased marketing	Software, prepackaged	Purchased marketing	Software, prepackaged	Purchased marketing
2008	6.5%	8.0%	0.9%	3.1%	0.06	0.25
2009	6.2%	2.7%	0.9%	3.4%	0.06	0.09
2010	4.0%	-0.5%	1.0%	3.6%	0.04	-0.02
2011	4.6%	0.9%	0.9%	3.5%	0.04	0.03
2012	9.9%	3.4%	0.9%	3.4%	0.09	0.12
2013	12.3%	5.1%	0.9%	3.5%	0.12	0.18
2014	11.5%	5.4%	1.0%	3.6%	0.11	0.19
2015	10.7%	5.4%	1.1%	3.6%	0.12	0.20
2016	11.2%	5.7%	1.1%	3.7%	0.13	0.21
2017	11.9%	6.0%	1.2%	3.9%	0.14	0.23
2018	12.9%	6.3%	1.2%	3.7%	0.16	0.23
2019	13.3%	6.8%	1.3%	3.5%	0.18	0.24
2020	12.8%	5.6%	1.4%	3.7%	0.19	0.21
1987-2020	17.9%	7.5%	0.8%	2.4%	0.10	0.15
1990-2000	26.9%	9.6%	0.5%	1.5%	0.11	0.14
2000-2007	11.6%	6.7%	0.9%	2.4%	0.10	0.16
2007-2020	10.3%	4.8%	1.1%	3.6%	0.11	0.17

 TABLE 4

 The Effect Of Prepackaged Software And Purchased Marketing On Output Growth

Notes: Reports the growth of stocks and the factor shares for two important elements of software and marketing, prepackaged software and purchased marketing. Though software stocks grow more rapidly, the factor share for purchased marketing is greater, and purchased marketing contributes more to output growth than prepackaged software does.

covered here represents about three-quarters of GDP, and excludes general government, government enterprises, nonprofits, and households.

Figure 3.a shows the flow of capital services to the goods sector and to the services sector over time. We calculate the annual flow of capital services in each industry and add them for all goods and for all services. Capital services were slightly less in goods from 1987 to 1990. However, by 2020 only 30 percent of capital services occurred in goods.

Figure 3.b shows that the intangible share of capital services was originally greater in goods than in services. It was not until about 2001 that the intangible share in services surpassed the share in goods, so that the expanding role of services began to increase the overall amount of intangibles. R&D represents a large portion of the intangibles in manufacturing. Figure 3.b shows that, at the end of the technology boom in 2000, the share of capital payments spent on intangibles began to decline for goods but continued to increase in services.

5. INFLUENCES ON MARKETING ASSETS

Our data on marketing in 61 industries make it possible to examine further issues, such as whether the amount of marketing varies between industries oriented towards consumer or business markets. Sections 5.1 and 5.2 find no consistent patterns, but we include these results for completeness.



14754991, 0, Downloaded from https://onlinelibrary.wiley.com/doi/101111/roiw.12678 by Boston University, Wiley Online Library on [15/102024], See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License





Figure 3. (a) Flow of Capital Services to the Production of Goods and Services, 1987–2020.
 Notes: It Reports the Total Amount of Capital Services, in Goods and in Services Industries, Over Time. (b) The Intangible Share of Capital Services, in Goods and in Services, 1987–2020. It shows the Share of Capital Services provided by Intangibles, Separately for Goods and Services Industries, Over Time

5.1. The impact of consumer or business markets on the level and growth of marketing

IO tables provide information on how much output of each commodity is delivered to intermediate products, consumption, or investment. The amounts used in consumption tell us how important the consumer market is, and amounts used in intermediate products and investment show how important the business market is in each of our 61 industries.¹⁸ We use data from BEA's detailed 2012 IO table.

We seek to understand how marketing practices differ between consumer and business-oriented industries. In the cross-section we measure the importance of marketing in each industry by the flow of capital services to marketing as a proportion of that industry's value added. We examine the growth of marketing investment and marketing's influence on labor productivity growth.

¹⁸Since some output is delivered to government, the shares of consumer and business are not perfectly collinear.

Published 2024. This article is a U.S. Government work and is in the public domain in the USA. *Review of Income and Wealth* published by John Wiley & Sons Ltd on behalf of International Association for Research in Income and Wealth.

We find no evidence that the intensity or rate of growth of marketing activities differs between industries oriented to consumer or business markets. The shares of purchased and own-account marketing similarly do not differ between consumer or business industries. Defining marketing intensity in industry *i* as the flow of capital services to marketing divided by the value added observed in that industry, we estimate the following regression:

(4) Marketing Intensity_i = $\alpha + \beta$ Consumer Share_i + γ Business Share_i

These regressions show no sign that the consumer or business orientation characteristic of an industry affects observed marketing intensity. Similarly, the type of customer does not affect the intensity of purchased and own-account marketing or our measures of time-series effects.

We had expected to find more marketing in consumer-oriented industries. The World Advertising Research Center (WARC) occasionally reports the U.S. industries in which advertising expenditures are the greatest. Their 2022 report lists these industries, in order, as retail, media and publications, business and industrial, financial services, technology and electronics, pharma and healthcare, technology and utilities, automotive, and amusement and leisure.¹⁹ That WARC list appears to be heavily weighted towards consumer goods.

A possible explanation is that national income measures assign a firm's advertising to each of its establishments, which are often classified in different industries. U.S. National Income and Product Accounts data report heavy advertising expenditures in wholesale trade, financial functions, and management of companies. Such measures probably assign advertising to economic functions well. These national income conventions may explain why we cannot establish a relationship between the customer type and observed marketing.

5.2. The effect of the presence of ICT on the future growth of marketing

We hypothesized that the presence of information and communication technology (ICT) would lead to a more rapid growth of investment in marketing, and that the link between ICT and the subsequent growth of marketing became stronger in more recent years, as digital marketing became more prevalent. We measured the presence of ICT in each industry in any year as the share of ICT assets, including software, in current value added.²⁰

We did not find any clear impact of ICT on marketing in our U.S. industry data. With more detailed data, such as information on many firms in the same industry, or data for the same sector in different countries (Chen et al., 2016), the effects of ICT might be clearer.

¹⁹https://www.marketingcharts.com/advertising-trends/spending-and-spenders-227936/7?et

<u>_blog</u>.The website Zippia has a somewhat similar list of the industries, but their list is restricted to digital advertising.

²⁰We consider investment in marketing from 1990, 2000, and 2007, the beginning of each of our three subperiods. We also consider growth since the expansion of digital marketing in 2012. For each of these four time periods, we cannot establish any effect of ICT intensity on the future growth of marketing investment.

5.3. The effect of advertising if digital advertising is substantially more effective

Mandel (2019) emphasizes that digital advertising, viewed on personal computers or mobile phones, is inherently more effective than print media advertising. Digital advertisers know more about the interests and concerns of potential customers and can target or customize ads towards likely buyers. This is a quality change, in the same sense that cars with more horsepower and houses with more square footage are of higher quality and represent more output. Consistent with that hypothesis, advertisers are shifting to digital advertising very rapidly. The Service Annual Survey shows that the digital share of the advertising market increased from 0.9 percent in 2002 to 38.2 percent in 2015 and 58.3 percent in 2020.²¹ Growth of this magnitude suggests that digital advertising offers important advantages to advertisers, most notably the targeting of specific consumers. As Mandel states (2019, p. 4) "In the economic sense, digital advertising is more productive than print advertising." Also (p. 12) "The simplest explanation for all these observations is that advertisers are finding that they can get a bigger bang for their buck by spending their money online rather than in print."

Mandel (2019) suggests that digital advertising is five-thirds as effective as print advertising. That is, every dollar spent on digital advertising brings a bonus of 0.67 cents of extra output due to the greater effectiveness of digital ads. With a 60 percent increase in the digital share over the years, that would imply 60 * 2/3 or a 40 percent increase in the effective amount of advertising just from the switch to the internet. That seems to be a remarkable amount of additional advertising output, even allowing for the overwhelming success of firms like Google, Facebook, and TikTok. Perhaps these magnitudes arise because Mandel was comparing digital advertising with print media, which is a particularly stagnant advertising category.

Even if the quality differences are not so large as Mandel suggests, it is plausible that typical deflators do not adequately adjust for quality improvements in advertising. To examine these possibilities, Table 5 considers effectiveness bonuses of 0.10 percent or 0.20 percent for every 1 percent increase in the digital share. In these cases, the long-term 60 percent increase in the digital share would be associated with a 6 or 12 percent gain in the real amount of advertising. These increases in output are strongest since 2015 when the digital share of advertising increased from 38 to 58 percent.

The first column of Table 5 shows the digital share of the advertising market from 2002 to 2020, from the SAS. The second and third columns report the extra bonus of advertising output if each additional dollar spent on digital advertising brings a bonus of 10 or 20 cents of additional output.

Table 5 shows that if digital advertising brings even modest productivity advantages, advertising output increases 6 to 12 percent by 2020 solely because of the shift to the internet. Equivalently, the price per unit of advertising output would decline by 6 or 12 percent by 2020 just because of the output expansion due to digital advertising. In 2020, the present official estimate of advertising output price, 103.696 declines as much as 10 percent, to 98.01 (103.696/1.058) or 92.92 (103.696/1.116).

²¹This digital share refers to the industry 5418 only. Purchases of the commodity advertising from other industries, purchases of other marketing services, or own-account marketing may have different digital shares.

Year	(1) Digital share	(2) = 10% of (1) Bonus, model 1	(3) = 20% of (1) Bonus, model 2	(4) BEA price
2002	0.9	0.1	0.2	82.636
2003	1.5	0.2	0.3	83.778
2004	2.5	0.3	0.5	84.609
2005	4.1	0.4	0.8	89.700
2006	7.1	0.7	1.4	95.709
2007	11.4	1.1	2.3	97.537
2008	13.6	1.4	2.7	98.088
2009	16.0	1.6	3.2	97.696
2010	18.5	1.9	3.7	97.612
2011	22.3	2.2	4.4	98.723
2012	26.0	2.6	5.2	100.000
2013	29.1	2.9	5.8	101.210
2014	30.8	3.1	6.2	102.127
2015	38.2	3.8	7.6	102.775
2016	42.1	4.2	8.4	103.337
2017	47.1	4.7	9.4	103.809
2018	50.7	5.1	10.0	103.952
2019	55.1	5.5	11.0	104.915
2020	58.3	5.8	11.6	103.696

TABLE 5 THE DIGITAL SHARE OF THE ADVERTISING MARKET AND THE DIGITAL BONUS, 2002–2020

Note: Estimates of the digital share of advertising are obtained from the Service Annual Survey. The digital share reported here is for the advertising industry. Digital shares for commodity advertising are broadly similar. Mandel mentioned to us that a smaller proportion of digital advertising is long-term investment, which tends to attenuate the productivity bonus.

These calculations show that the implied effect on the price of advertising is substantial even if digital ads are only slightly more effective than other forms of advertising. We do not at present know exactly how much more effective digital ads are. However, this exercise has shown that, even if digital ads were only slightly more effective, that is sufficient to lower the implied price of advertising substantially. Lower prices would in turn show that advertising has increased output growth more rapidly. Further research on the productivity advantage of digital advertising would be helpful.²²

6. DISTRIBUTION OF ASSETS ACROSS INDUSTRIES AND THEIR EFFECT ON GROWTH

6.1. Stocks of asset types in different industries

We now consider the importance of asset types at industry level. The sectors considered are manufacturing, other goods, trade, finance, and other services. Table 6 shows the importance of each type of capital as a percentage of total

²²If digital marketing represents unmeasured quality improvement, then marketing prices would have increased less than currently available price indices indicate, so that the amount of output and the productivity gains associated with marketing would be greater than presently measured. Another crucial question about advertising or marketing is whether expenditures by one firm cancel expenditures by a rival firm, so that the net effect of expenditures is reduced. In our judgment, this topic cannot be understood solely from industry data. As in the analysis of R&D (Bloom et al., 2013), this issue requires both firm and industry data, so that it is possible to evaluate the presence of positive or negative spillovers.

		Equipmen	it Stri	uctures	Inventories		Land	Total tangible
Manufacturing		30.6%		2.4%	5.5%		25.0%	73.5%
Other goods		16.8%		3.7%	23.8%		54.9%	99.3%
Trade		14.8%	0	3.9%	22.3%		32.9%	93.8%
Finance, insurance	, real estate	14.5%		0.2%	21.0%		61.0%	96.7%
Other services		22.1%		1.5%	13.2%		44.1%	80.8%
	Purchased	Own acct	Entertainment		Software,	Software,	Software,	Total
	marketing	marketing	originals	R&D	custom	own acct	prepackaged	intangible
Manufacturing	1.0%	0.2%	0.0%	23.6%	0.9%	0.3%	0.4%	26.5%
Other goods	0.2%	0.0%	0.0%	0.3%	0.1%	0.0%	0.1%	0.7%
Trade	3.2%	0.4%	0.0%	1.0%	0.8%	0.4%	0.5%	6.2%
FIRE	1.2%	0.2%	0.0%	0.3%	0.7%	0.2%	0.7%	3.3%
Other services	3.2%	0.7%	6.7%	5.0%	2.0%	1.0%	0.7%	19.2%

TABLE 6

· ·	4.5			
- 1	1 0.3		5 1	
-			N	

capital stocks in each of these five sectors. Panel A of Table 6 reports tangible assets and panel B shows intangible assets for 2012. Equipment accounted for 31 percent of capital stocks in manufacturing. Similarly, inventories were 24 percent of all stocks in trade. In panel B, we see that R&D accounts for 24 percent of all manufacturing capital stocks and entertainment originals are 7 percent of total stocks in other services. Purchased marketing is most important in trade and other services, accounting for 3 percent of total stocks.

6.2. Correlations between sources of growth

Our breakdown attributes the growth of output to 14 inputs and a TFP residual. We measure these effects for each of our 61 industries. The 14 inputs shown in Table 7 are: 7 types of intangibles, five types of tangible capital, labor hours, and labor composition. These measures are expressed as average annual contributions over the 1987–2020 period.

Table 7 shows how the average annual contributions to growth are correlated across industries. We report correlations of special interest in bold type. The high correlations between various forms of software show that industries which use one form of software tend to use others as well. The contributions of the two forms of marketing are highly correlated. Both forms of marketing are also highly correlated with the impact of software. Own-account marketing is largely measured by the presence of computer-oriented occupations, so that a link with software is not surprising. However, purchased marketing, which consists largely of advertising, is also closely linked to the presence of software.

There is some support for the well-established connection between R&D and marketing, especially for own-account marketing (Corrado & Hao, 2014). We had thought that potential drivers of economic growth such as ICT, improvements in the composition of labor, or TFP might be associated with a more rapid growth of intangibles. There is some evidence that ICT (which here excludes software) may have some effect on the growth of intangibles, but measures of labor composition and TFP appear to have little connection to intangibles growth. Our estimates of TFP may be subject to measurement error, partially because they are based on value added rather than gross output.²³

14754991, 0, Downloaded from https://onlinelibrary.wiley.com/doi/101111/roiw.12678 by Boston University, Wiley Online Library on [15/102024], See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

The economics literature, such as Bessen et al. (2020), discusses how intangibles have altered the nature of production, based on firm data. This literature typically concentrates on firm data because intangibles frequently affect firms in the same industry differently (De Ridder, 2024). However, Table 7 shows that differences between industries can also describe some of the connections between intangibles.

²³Berndt and Wood (1975) suggest that measures of productivity growth based on gross output are preferable to estimates obtained from value added. De Loecker and Scott (2016) and Gandhi et al. (2020) show the usefulness of gross output methods and describe how they can be implemented. Connections between TFP and the presence of intangibles might provide some clues about which intangibles have spillover effects. For example, there is a mild positive connection between R&D and TFP. However, industries differ in so many other respects that it is preferable to search for spillovers within data for firms in a single industry.

		osition													.08	ces in 61 patterns 1 certain interest.
	Labor	Comp													0	influen ustrate untity ii special
s		Hours												-0.27	-0.20	ion of 15 ries, to ill sured que ich are of
USTRIE	ng	Own acct											0.16	0.13	-0.05	ntributi 1 indust ro mea ons whi
cross 61 Ind	Marketi	Purch-ased										0.75	0.18	0.05	-0.13	atrix of the co is across the 6 eting—have ze icates correlati
Зкомтн, Ао		Ent. Originals									0.17	0.05	0.00	-0.08	-0.04	provides a m contribution count Marke als Bold indi
-2020 0		R&D								-0.04	0.25	0.36	-0.04	0.16	0.17	0. That _I ch of the Own Ac d numera
ons To 1987		Inventories							0.22	-0.17	0.10	-0.03	-0.11	0.02	0.14	n 1987 to 202 n between ead riginals, and text are in bol
TRIBUTIC		Land						-0.07	0.16	-0.01	0.38	0.50	0.03	0.14	-0.03	astry fror correlatio inment O ed in the 1
TABLE 7 VED BY CON		Structures					0.18	-0.10	-0.04	0.35	0.10	0.15	0.15	-0.21	-0.11	in every indu e reports the c &D, Entertai ons emphasize
th, Measuf		Own acct				0.14	0.49	-0.04	0.35	-0.03	0.71	0.94	0.17	0.08	-0.08	atput growth ts.) This table in inputs—R es. Correlatio
DF Grow	Software	Custom			0.96	0.13	0.42	-0.02	0.42	-0.03	0.73	0.94	0.15	0.12	0.00	ution to ou ne 14 inpu ced. Certai se industri
N SOURCES (01	Purchased		0.81	0.71	0.17	0.29	-0.03	0.28	-0.04	0.67	0.75	0.14	0.12	-0.04	nnual contrib addition to tl losely associal output in the
Betwee	ent	ICT	020	09.0	0.59	0.20	0.31	-0.08	0.23	0.09	0.64	0.63	0.14	-0.09	-0.13	iverage al luence in latively cl oution to
ORRELATIONS	Equipme	Except ICT	0.43	-0.02	-0.02	0.14	-0.05	0.04	-0.04	-0.06	0.12	-0.04	0.03	-0.24	-0.25	each factor's a ed as a 15th inf output are re- nake no contrib
C			ICT Dumband antimation	Custom software	Own acct software	Structures	Land	Inventories	R&D	Ent. originals	Purchased marketing	Own acct marketing	Labor hours	Labor composition	TFP residual	<i>Notes:</i> We measure industries. (TFP is include in which contributions to industries and therefore m

	THAT INTANGIBLE IN THE I	RIVATE SECTOR	
Intangible	1987 industry concentration	2002 industry concentration	2020 industry concentration
Purchased marketing	66%	62%	70%
Own account marketing	69%	70%	74%
Entertainment originals	100%	100%	100%
R&D	82%	76%	79%
Software, custom	66%	67%	80%
Software, own account	66%	72%	84%
Software, prepackaged	63%	63%	70%
Total marketing	63%	62%	69%
Total software	66%	66%	75%

TABLE 8 STOCKS OF EACH INTANGIBLE IN THE TOP 10 INDUSTRIES, AS A PERCENTAGE OF THE TOTAL STOCK OF THAT INTANGIBLE IN THE PRIVATE SECTOR

Notes: Reports the share of each intangible stock that was held by the top 10 industries. Data are reported for 1987, 2002, and 2020. For each intangible, the top ten means the ten industries with the greatest stock of that asset.

6.3. Concentration of intangibles among industries

This subsection provides evidence on the extent to which the use of each intangible is concentrated in a few leading industries. Table 8 reports industry concentration for each intangible in 1987, 2002, and 2020, as measured by the percentage of the total stock of that intangible observed in the top 10 of our 61 industries.

Entertainment originals always have concentration of 100 percent, since only five industries hold this asset. Concentration of R&D declines modestly. However, concentration of software and marketing increases markedly, especially after 2002. Bessen (2022) describes how software has become more proprietary since 2000, as firms develop their own computer systems.

Much of the concentration of software has occurred within industries, as firms with effective digital systems displace their competitors. However, Table 8 shows that, since 2002, software also became more concentrated across industries; each type of software also became more concentrated. Both forms of marketing similarly became more concentrated since 2002. The same internal data systems that are known to make the software systems of leading firms more effective are likely to make the same firms' marketing more successful and concentrated.

Table 9 lists the five industries with the largest stocks of purchased and own-account marketing, in order, in 1987, 2002, and 2020. The lists of leading industries are generally reasonable. Conventional lists of leading advertisers are likely to emphasize consumer industries, such as retail trade, pharmaceuticals, electronics, automotive, food, and finance. National accounts methods frequently assign advertising expenditures to different functions of a firm, such as retail or wholesale trade, finance, or the management of companies, rather than to the

Purchased marketingPurchased marketingRetail tradeWholesale tradeWholesale tradeWholesale tradeConstructionBroadcasting and telecommunicationsManagement of companies andFood services and drinking placesFood services and drinking placesFood services and drinking placesFood services and drinking placesFood services and drinkingManagement of companies andFood services and drinking placesPublishing andBroadcasting andBroadcasting andBroadcasting andInternet (includes software)Internet (includes software)Management of companies andInternet (includes software)Molesale tradeWholesale tradePublishing industries, exceptBroadcasting andInternet (includes software)Computer and electronicComputer and electronic <th>2002 2020</th>	2002 2020
Windlesale tradeWindlesale tradeConstructionBroadcasting and telecommunicationsManagement of companies and enterprisesFood services and drinking placesManagement of companies and enterprisesMiscellaneous professional, scientific, and technical servicesFood services and drinking placesMiscellaneous professional, scientific, and technical servicesManagement of companies and telecommunicationsManagement of companies and enterprisesManagement of companies and tinternet (includes software)Management of companies and enterprisesManagement of companies and telecommunicationsManagement of companies and publishing industries, except internet (includes software)Commuter and electronic telecommunicationsPublishing industries, except telecommunications	Retail trade
ConstructionBroadcasting and telecommunicationsManagement of companies and enterprisesFood services and drinking placesManagement of companies and enterprisesFood services and drinking placesFood services and drinking placesMiscellaneous professional, scientific, and technical servicesPlacesManagement of companies a telecommunicationsData niternet (includes software)Broadcasting and enterprisesManagement of companies and telecommunicationsManagement of companies a enterprisesManagement of companies and telecommunicationsManagement of companies a enterprisesManagement of companies and telecommunicationsManagement of companies a enterprisesManagement of companies and telecommunicationsManagement of companies and enterprisesManagement of companies and telecommunicationsManagement of companies and enterprisesManagement of companies and telecommunicationsManagement of companies and publishing industries, exceptMolesale trade internet (includes software)Publishing industries, exceptCommuter and electronic tomuter and electronicPublishing industries, exceptCommuter and electronicCommuter and electronic	Wholesale trade
Management of companies and enterprisesFood services and drinking placesFood services and servicesMiscellaneous professional, scientific, and technical servicesFood services and drinking placesMiscellaneous professional, scientific, and technical servicesOwn-account marketing Broadcasting and telecommunicationsManagement of companies a enterprisesOwn-account marketing Broadcasting and telecommunicationsManagement of companies a enterprisesManagement of companies and internet (includes software)Management of companies and publishing, and other information servicesManagement of companies and enterprisesData processing, internet publishing, and other information servicesMolesale trade internet (includes software)Publishing industries, except internet (includes software)Computer and electronic internet (includes software)Publishing industries, except internet (includes software)	Tederal Reserve banks, credit intermediation, and related
Fourtering placesMiscellaneous professional, scientific, and technical services placesOwn-account marketing Own-account marketing 	aces Miscellaneous professional, scientific and technical service
CommunicationsBroadcasting and Broadcasting andManagement of companies a enterprisesBroadcasting and Broadcasting and 	cientific, and technical services Management of companies and enterprises
telecommunicationstelecommunicationsenterprisesPublishing industries, exceptManagement of companies andData processing, internetinternet (includes software)enterprisesminformation servicesManagement of companies andWholesale tradeWholesale tradeManagement of companies andWholesale tradewholesale tradeMolesale tradePublishing industries, exceptBroadcasting andInternet (includes software)internet (includes software)telecommunications	Management of companies and
Publishing industries, exceptManagement of companies and internet (includes software)Data processing, internet and other information servicesManagement of companies and 	enterprises
Management of companies and Wholesale trade Information services Management of companies and Wholesale trade Wholesale trade enterprises Publishing industries, except Broadcasting and Wholesale trade internet (includes software) telecommunications Commuter and electronic Commuter and electronic Publishing industries, except	nd Data processing, internet publishing, and other
Wholesale trade Publishing industries, except Broadcasting and internet (includes software) telecommunications Computer and electronic Computer and electronic secont	mormation services Wholesale trade
Computer and electronic Computer and electronic Publishing industries, except	Broadcasting and telecommunications
products products internet (includes software	Publishing industries, except internet (includes software)

TABLE 9

final product eventually sold. That probably explains why relatively few consumer industries appear on the list of the largest marketers.²⁴

7. Conclusions

The IMF report on marketing assets urged examination of the feasibility of incorporating these assets into the national accounts. The summary of the IMF report on marketing assets (2022, p. 2) states "As part of the global consultation, it is proposed to enquire to what extent economies still face measurement challenges, which prevent capitalizing marketing assets." This paper shows that, for the United States, building on the approach of Corrado et al. (2005, 2009) and Corrado & Hao (2014), this is feasible.

It would be useful to develop an understanding of how sales workers affect the value of marketing assets and to integrate emerging work on the value of trademarks (Dosi et al., 2022) with a more general view of firm assets and capabilities. Two trends emerging in the literature are likely to have a strong impact on how marketing and other intangibles are understood. Mandel (2019) has argued that it is difficult to develop measures of the output price of marketing because recent shifts to digital advertising, such as through the internet or smart phones, make marketing far more productive than prior advertising media. Koh et al. (2020) propose that a portion of the added income produced by capitalization of intangibles should be credited to labor rather than to capital. These two lines of thought could have a strong influence on how marketing is eventually understood. In addition, time diaries and other surveys could help estimate what proportion of time advertising and own-account marketing workers devote to long-term investment. Finally, there are also further issues to consider, such as the appropriate treatment of licensing and franchising.

Despite these topics that require further attention, the clear message of our paper is that a remarkable amount can be done to develop a comprehensive treatment of marketing for the United States. It is possible to construct measures of purchased advertising, other purchases of marketing services, and own-account marketing. These measures rest on solid and highly detailed data. Overall, the results of our study strongly suggest that many statistical agencies will be able to include marketing capital in their accounts.

In particular, the summary paragraph of the initial IMF discussion of marketing assets (IMF, 2022) states that "the major reason for not treating marketing assets as fixed assets is due to the difficulty of measuring their value." Note that the procedures developed in this paper provide rental prices as well as stocks of marketing capital, and therefore overcome the main difficulty that the IMF document emphasizes.

Our most central empirical result is that marketing contributes about as much to overall output growth as R&D or software. Table 1 shows that the contribution

²⁴It is probably an anomaly that construction is the third largest purchaser of advertising in 1987. Recall that, prior to 1997, we measure advertising from data on "miscellaneous professional, scientific, and technical services." This category includes items such as accounting, architectural, engineering, and design services as well as advertising. Shifts among these categories could overstate advertising in construction in 1987.

of marketing to output growth increased over time, whereas the contributions of R&D and software tended to stabilize. Own-account marketing grew more quickly than purchased marketing, steadily over the entire period. Purchases from web design and hosting and from marketing services, together with increased own-account employment of technical and marketing skills, all helped to drive marketing investment. However, capitalization of both forms of marketing has only a modest effect on the growth of output.

These estimates of the impact of marketing largely occur because we have classified certain elements of the revolution in computers and data as contributors to marketing. Investments in web design and hosting are certainly a central element of marketing investment. Similarly, as in Corrado & Hao (2014), own-account investments in computer and marketing occupations are crucial in developing the internal capabilities of the firms that comprise each industry. It will be important to determine how these investments should be allocated between the data revolution and marketing.

References

- Bagwell, K. (2005). The economic analysis of advertising, Columbia University Department of economics, Discussion Paper No. 0506-01, 2005.
- Berndt, E., & Wood, D. (1975). Technology, prices, and the derived demand for energy. *Review of Economics and Statistics*, 57, 259–68.
- Bessen, J. (2020). Industry concentration and information technology. *Journal of Law and Economics*, 63, 531–55.
- Bessen, J. (2022). The new goliaths: How corporations use software to dominate industries, kill innovation, and undermine regulation. Yale University Press.
- Bessen, J., Denk, E., Kim, J., & Righi, C. (2020). Declining Industrial Disruption, Boston University School of Law, Law and Economics Working Paper 20–28. Also, Social Science Research Network. https://ssrn.com/abstract=3682745
- Bloom, N., Schankerman, M., & Van Reenen, J. (2013). Identifying technology spillovers and product market rivalry. *Econometrica*, 81, 1347–93.
- Bronnenberg, B., Dube, J., & Gentzkow, M. (2012). The evolution of brand preferences: Evidence from consumer migration. *American Economic Review*, 102, 2412–508.
- Bursztyn, L., & Cantoni, D. (2016). A tear in the iron curtain: The impact of Western television on consumption behavior. *Review of Economics and Statistics*, 98, 25–41.
- Chen, W., Niebel, T., & Saam, M. (2016). Are intangibles more productive in ICT-intensive industries? Evidence from EU Countries. *Telecommunications Policy*, 40, 471–84.
- Corrado, C., & Hao, J. (2014). Brands as productive assets: Concepts, measurement, and global trends," WIPO Economic Research Working Paper No. 13.
- Corrado, C., Haskel, J., Jona-Lasinio, C., & Iomma, M. (2022a). Intangible capital and modern economies. *Journal of Economic Perspectives*, 36, 3–28.
- Corrado, C., Haskel, J., Jona-Lasinio, C., Iomma, M., & Bontadini, F. 2022b. Data, intangible capital, and productivity, (conference presentation), 10th IMF statistical forum: Measuring the tangible benefits of intangible capital.
- Corrado, C., Hulten, C., & Sichel, D. (2005). Measuring capital and technology: An expanded framework. In C. Corrado, J. Haltiwanger, & D. Sichel (Eds.), *Measuring Capital in the new Economy, studies in income and wealth* (Vol. 65, pp. 11–46). University of Chicago Press.
- Corrado, C., Hulten, C., & Sichel, D. (2009). Intangible capital and U.S. economic growth. *Review of Income and Wealth*, 55, 661–85.
- De Loecker, J., & Scott, P. (2016). Estimating Market Power: Evidence from the U.S. Beer Industry, National Bureau of Economic Research Working Paper 22957.
- De Ridder, M. (2024). Market power and innovation in the intangible economy. American Economic Review, 114, 199–251.
- Dosi, R., Gavrilova, E., Silva, R., & Soares, M. (2022). The Value of Trademarks, Social Science Research Network. https://ssrn.com/abstract=4280505

- Fraumeni, B., & Okubo, S. (2005). R&D in the National Income and product accounts: A first look at the effect on GDP. In C. Corrado, J. Haltiwanger, & D. Sichel (Eds.), *Measuring Capital in the new Economy, studies in income and wealth* (Vol. 65, pp. 275–322). University of Chicago Press.
- Gandhi, A., Navarro, S., & Rivers, D. (2020). On the identification of gross output production functions. *Journal of Political Economy*, 128, 2973–3016.
- Heys, R., & Fotopoulou, E. 2022. Delivering tangible estimates of non-SNA intangible assets, (conference presentation), 10th IMF statistical forum: Measuring the tangible benefits of intangible capital.
- International Monetary Fund. (2022). Committee on Balance of Payments Statistics BPM6/2008/SNA Update, Joint Globalization Task Force (GZTT), Inter-Secretariat Working Group on National Accounts, Section G9, Guidance Note on Marketing Assets, Payments for Nonproduced Knowledge-Based Capital (Marketing Assets. https://unstats.un.org/unsd/nationalaccount /SNAUpdate/GZTT.asp
- International Monetary Fund. (2023). SNA/M1.23/7 Summary of Discussions for the Joint AEG-BOPCOM Meeting. https://unstats.un.org/unsd/nationalaccount/aeg/2023/M22/M22 &uscore;7&uscore;Summary&uscore;of&uscore;Discussions.pdf
- Koh, D., Santaeulalia, R., & Zheng, Y. (2020). Labor share decline and intellectual property products capital. *Econometrica*, 88, 2609–28.
- Lodish, L., Abraham, M., Livelsberger, J., Lubetkin, B., Richardson, B., & Stevens, M. (1995). A summary of fifty-five in-market experimental estimates of the long-term effect of TV advertising. *Marketing Science*, 14, G133–40.
- Lucking, B., Bloom, N., & Van Reenen, J. (2019). Have R&D spillovers declined in the 21st century? *Fiscal Studies*, 40, 561–90.
- Mandel, M. (2019). The declining cost of advertising: Policy implications. Progressive Policy Institute.
- Martin, R., Mohnen, P., Thomas, C., Verhoeven, D., & Guillard, C. (2022). Efficient Industrial Policy for Innovation: Standing on the Shoulders of Hidden Giants, Social Science Research Network. https://ssrn.com/abstract=4026731
- Moyer, B., Planting, M., Fahim-Nader, M., & Lum, S. (2004). Preview of the comprehensive revision of the annual industry accounts. Survey of Current Business, 84, 38–51.
- Nakamura, L., Samuels, J., & Soloveichik, R. 2017. Measuring the "free" digital economy within the GDP and productivity accounts. Federal Reserve Bank of Philadelphia Working Paper 17-37.
- Ribarsky, J. (2022). Intangible Capital in the Update of the international Statistical Standards, (conference presentation). In 10th IMF International Statistical Forum: Measuring the Tangible Benefits of Intangible Capital.
- Rogers, R., & Tokle, R. (1995). The economics of advertising: Where's the data? *Review of Industrial* Organization, 10, 675–87.
- Silk, A., & Berndt, E. R. (2020). Aggregate Advertising Expenditure in the U.S. Economy: What's Up? Is it Real?", National Bureau of Economic Research Working Paper 28161.
- Soloveichik, R., 2013. Long-Lived Television Programs as Capital Assets,= Unpublished paper.
- Stiroh, K. (2002). Are IT spillovers driving the new economy? American Economic Review, 92, 1559-76.
- Sveikauskas, L. (2007). R&D and Productivity Growth: A Review of the Literature, Bureau of Labor Statistics Working Paper Number 408.
- U.S. Bureau of Economic Analysis. (2021). NIPA Handbook: Concepts and methods of the U.S. National Income and Product Accounts.
- Villalonga, B. (2004). Intangible resources, Tobin's q, and sustainability of performance differences. Journal of Economic Behavior and Organization, 54, 205–30.
- Vitorino, M. A. (2014). The effect of advertising on stock returns and firm value: Theory and evidence from a structural model. *Management Science*, 60, 227–45.

SUPPORTING INFORMATION

Additional supporting information may be found in the online version of this article at the publisher's web site:

Data S1 Supporting Information. **Data S2** Supporting Information.